Author’s response to reviews

Title: Tigecycline-induced sustained and severe hypoglycemia: a case report

Authors:

Yixin Chen (yixin29@zju.edu.cn)
Lin Li (3312012@zju.edu.cn)
Nan Zhang (3198047@zju.edu.cn)
Hong Li (srrshnfm@zju.edu.cn)

Version: 2 Date: 13 May 2019

Author’s response to reviews:

Dear, Editors:

Thank you very much for having our manuscript (ID: PHAT-D-19-00024R1) entitled “Tigecycline-induced sustained and severe hypoglycemia: a case report” reviewed in a timely and professional manner and for giving us an opportunity to revise the manuscript. We also deeply appreciate the reviewers for their critical review of the manuscript with thoughtful and constructive comments, based on which we have revised the manuscript. While the changes have been made in the revised manuscript, our point-by-point responses to the reviewers’ comments are detailed below.

Responds to the reviewers’ comments:

Reviewer 1:
1) We have altered the title, by replacing "sustaining" with "sustained".
2) We have replaced "over secretion" with "oversecretion" in the abstract and conclusion sections.
3) "Over production" was replaced with "overproduction" in the discussion section.
4) We have replaced "for discussing" with "to discuss" in the case presentation section.
5) Figure 1-5 axes are labelled with "blood glucose (mmol/L)".

Best Regards.
Yours sincerely,
Hong, Li
Institute: Zhejiang University School of Medicine Sir Run Run Shaw Hospital, Endocrinology
Address: east Qing chun road No.3, Hangzhou, China, CN 310016,
Email: srrshnfm@zju.edu.cn
Tel: 13588706366